Author's response to reviews

Title: Tactile massage and hypnosis as a health promotion for nurses in emergency care - a qualitative study

Authors:

Fanny Airosa (fanny.airosa@karolinska.se)
Susanne K Andersson (susanne.k.andersson@ki.se)
Torkel Falkenberg (torkel.falkenberg@ki.se)
Christina Forsberg (christina.forsberg@ki.se)
Elisabeth Nordby-Hörnell (enordbyhornell@telia.com)
Gunnar Öhlén (gunnar.ohlen@karolinska.se)
Tobias Sundberg (tobias.sundberg@ki.se)

Version: 3 Date: 14 September 2011

Author's response to reviews: see over
Dear Editor,

Here is a point-by-point response to each of the concerns of the manuscript "Tactile massage and hypnosis as a health promotion for nurses in emergency care – a qualitative study".

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)

Reply: The text has been revised accordingly.

Author list: Please restrict equal contributions to only two authors on the author list.

Reply: The text has been revised accordingly.

Abstract: This should be identical in the manuscript and the submission system

Reply: The text has been revised accordingly.

References: Please provide functioning link to http://inuti.karolinska.se/templates/page_100041.aspx

Reply: The text has been revised accordingly.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove from the manuscript.

Reply: The text has been revised accordingly.

Tables: Please remove vertical lines.

Reply: The text has been revised accordingly.

Best regards,

Fanny Airosa

On behalf of the team of authors:

Fanny Airosa, Susanne Andersson, Torkel Falkenberg, Christina Forsberg, Elisabeth Nordby-Hörnell, Gunnar Öhlén, Tobias Sundberg.